

November 06,2014.

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 14339-review.doc).

Title: Non-Surgical Periodontal Therapy - An Update on Current Evidence

Author: Dr. Rahul S Bhansali

Name of Journal: *World Journal of Stomatology*

ESPS Manuscript NO: 14339

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) Dear Sir, the manuscript singles out Vector[®] system as during literature search it was found to be the only ultrasonic scaler brand that incorporates features that enable the linear oscillations to be directed in a parallel direction to root surface. It was developed as a better alternative to existing piezoelectric scalers in terms of patient compliance. Considering its mode of action it can be assumed that other brands with similar movement of scaler tip in respect to the root surface may result in similar clinical efficacy. There must be other brands available with similar mode of action but literature search did not produce any data regarding the same.

.....

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,



06/11/2014

Dr. Rahul S. Bhansali

MDS, Reader,

Department of Periodontology and Implantology,

Shri Guru Gobind Singh College of Dental Sciences & Research Center, Lalbagh Road,

Burhanpur - 450331, M.P., INDIA.

Email: drrahulbhansali@yahoo.co.in

Telephone: +919422278157